Author’s response to reviews

Title: Preschool children in out-of-hours primary care - a questionnaire-based cross-sectional study of factors related to the medical relevance of health problems

Authors:
Grete Moth (moth@ph.au.dk)
Linda Huibers (huibers@ph.au.dk)
Astrid Ovesen (alo@ph.au.dk)
Morten Bondo Christensen (mbc@ph.au.dk)
Peter Vedsted (p.vedsted@ph.au.dk)

Version: 2 Date: 07 Dec 2017

Author’s response to reviews:

Dear editor

Thank you again for considering our manuscript for publication.
Below you find a point-by-point response to the issues mentioned.

1. In the section concerning ethical approval we have added a sentence explaining that the participating GPs and patients were informed to consent to participate by filling out and returning the questionnaire.
2. In the section concerning ethical approval we have added a link to the official Danish website explaining why this particular study is not included in the type of studies that need ethical approval.
3. The heading of the manuscript have been revised accordingly.
4. We have elaborated further on the Funding section to make it clear that the funding bodies were not involved in carrying out the study nor in writing the manuscript.
5. A clean version of the manuscript has been made.